Author's response to reviews

Title: Stratification of the severity of critically ill patients with classification trees

Authors:

Javier Trujillano (jtruji@cmb.udl.es)
Mariona Badia (marionabadia@wanadoo.es)
Luis Servia (lservia@telefonica.net)
Jaume March (jaume.march@cmb.udl.cat)
Angel Rodriguez-Pozo (rodriguez@medicina.udl.cat)

Version: 6 Date: 29 November 2009

Author's response to reviews:

The BioMed Central Production Team.

Dear,

We have revised the manuscript according to the indications of The BioMed Central Production Team.

Major corrections:
The affiliations section has been corrected.
A new paragraph has been added to the authors' contributions section.
Reference 35 has been cited in the main text (shown before as reference 34).

Minor corrections:
The e-mail addresses of all authors have been added.
The visible vertical lines have been removed from the tables.
Titles have been removed from the figures.
The typography of the manuscript has been revised.

Yours sincerely,

Javier Trujillano
jtruji@cmb.udl.es